Author's response to reviews

Title: Effectiveness of endometrial resection transcervical hysteroscopic based preventive treatment on the recurrence of endometrial polyps in post-menopausal women.

Authors:

Jesus S Jimenez-Lopez (jjimenez.hdoc@salud.madrid.org)
Ana Granados-Sanmiguel (angranado@gmail.com)
Alvaro Tejerizo-Garcia (tejerizo@hotmail.com)
Jose L Muñoz-Gonzalez (jilmunozg@yahoo.es)
Gregorio Lopez-Gonzalez (goyolopez2@hotmail.co)

Version: 6 Date: 26 October 2014

Author's response to reviews: see over
Editor's Response to Comments:

I do not understand the abbreviations which are used. Please define all abbreviations HUA eg, TMX, HSC.

All abbreviations have been deleted in the text.

The term post menopausal metrorrhagia is unfamiliar to me. Bleeding after menopause is defined as post menopausal bleeding. The FIGO classification of abnormal uterine bleeding has excluded the term "metrorrhagia".

In the text we deleted the term metrorrhagia, we have replaced post menopausal bleeding

If a woman did not have any vaginal, why did she have any available sonographic examination? This is not clear.

Patients as part of clinical examination were subjected to transvaginal sonography, in some cases as part of the monitoring protocol when using tamoxifen treatment.

I did not receive a Figure 1 s can not comment.

In revising the text, we have decided to remove Figure 1 and Figure 2.

I presume Page 8 is the start of the Discussion but there is no heading.

We have the results section of the discussion following the directions of the publisher.

Table I is unclear. What is IMC?

We replaced by BMI: Body mass index

Table II. What is the meaning of Meses? This is not a term I am familiar with which.

We have replaced its English translation (month)

Table III. Can not interpret. No explanation and what is TMX?

TMX: tamoxifen

Table IV No explanation and I am uncertain what is being presented.

We have included the following explanatory text in the table.

Table 4: Level of satisfaction and acceptability of endometrial resection in postmenopausal patients. (Yes. The patient considers proper treatment and is pleased with the results. I recommend it to other patients.)
No. The patient does not consider the appropriate treatment and not satisfied with the results. And is not recommended to other patients.
N/C. The patient does not have any strong opinion.